Author's response to reviews

Title: Study Protocol: Phase III single-blinded fast-track pragmatic randomised controlled trial of a complex intervention for breathlessness in advanced disease.

Authors:

Morag C Farquhar (mcf22@medschl.cam.ac.uk)
A Toby Prevost (toby.prevost@kcl.ac.uk)
Paul McCrone (p.mccrone@iop.kcl.ac.uk)
Irene J Higginson (irene.higginson@kcl.ac.uk)
Jennifer Gray (pauline.kleanthous@addenbrookes.nhs.uk)
Barbara Brafman-Kennedy (barbara.brafman-kennedy@addenbrookes.nhs.uk)
Sara Booth (sara.booth@addenbrookes.nhs.uk)

Version: 3 Date: 4 May 2011

Author's response to reviews: see over
Dear Editor

Re: Manuscript 4101855505298504
Study Protocol: Phase III single-blinded fast-track pragmatic randomised controlled trial of a complex intervention for breathlessness in advanced disease.

Further to your email of 3rd May 2011, please find enclosed Manuscript 4101855505298504 with the corrections you requested: a figure title and legend section has now been included and the ethics and funding documents removed.

Yours sincerely

Dr Morag Farquhar
Macmillan Post-Doctoral Research Fellow